



Article Key Factors Influencing Design Learners' Behavioral Intention in Human-AI Collaboration Within the Educational Metaverse

Ronghui Wu ¹, Lin Gao ², Jiaxin Li ², Qianghong Huang ¹, and Younghwan Pan ^{1,*}

- ¹ Department of Smart Experience Design, Kookmin University, Seoul 02707, Republic of Korea; wuronghui@kookmin.ac.kr (R.W.); huangqianghong@kookmin.ac.kr (Q.H.)
- ² Department of Animation, Zhongyuan University of Technology, Zhengzhou 450007, China; 3363@zut.edu.cn (L.G.); 2023111511@zut.edu.cn (J.L.)

* Correspondence: peterpan@kookmin.ac.kr

Abstract: This study investigates the key factors which influence design learners' behavioral intention to collaborate with AI in the educational metaverse (EMH-AIc). Engaging design learners in EMH-AIc enhances learning efficiency, personalizes learning experiences, and supports equitable and sustainable design education. However, limited research has focused on these influencing factors, leading to a lack of theoretical grounding for user behavior in this context. Drawing on social cognitive theory (SCT), this study constructs a three-dimensional theoretical model comprising the external environment, individual cognition, and behavior, validated within an EMH-AIc setting. By using Spatial.io's Apache Art Studio as the experimental platform and analyzing data from 533 design learners with SPSS 27.0, SmartPLS 4.0, and partial least squares structural equation modeling (PLS-SEM), this study identifies those rewards, teacher support, and facilitating conditions in the external environment, with self-efficacy, outcome expectation, and trust in cognition also significantly influencing behavioral intention. Additionally, individual cognition mediates the relationship between the external environment and behavioral intention. This study not only extends SCT application within the educational metaverse but also provides actionable insights for optimizing design learning experiences, contributing to the sustainable development of design education.

Keywords: educational metaverse; sustainable design education; human-AI collaboration; social cognitive theory; behavioral intention

1. Introduction

Educational equity is a core issue of global sustainable development [1]. The United Nations Sustainable Development Goals (SDGs) point out that ensuring inclusive and equitable quality education and promoting lifelong learning for all are the core tasks of global education [2]. In design education, geographical barriers, uneven resources, and lack of opportunities for interdisciplinary collaboration pose serious challenges to educational equity. This imbalance restricts the sustainable development of design education, especially for students with limited resources who find it difficult to obtain the same learning opportunities and support as other students [3,4]. With the widespread application of emerging technologies such as virtual reality (VR), augmented reality (AR), and artificial intelligence (AI) in the design industry, the problem of uneven distribution of traditional educational resources has become more prominent, placing higher demands on the rapid transformation of design education. Therefore, this study aims to explore how to effectively alleviate educational inequality by applying artificial intelligence and the educational metaverse, thus promoting the sustainable development of design education.

The rise of the educational metaverse provides a new opportunity to solve the unfair problems encountered in design education. The educational metaverse breaks through the geographical and resource limitations of traditional education through immersive and personalized learning experiences [5–7]. For example, Google Expeditions allows



Citation: Wu, R.; Gao, L.; Li, J.; Huang, Q.; Pan, Y. Key Factors Influencing Design Learners' Behavioral Intention in Human-AI Collaboration Within the Educational Metaverse. *Sustainability* **2024**, *16*, 9942. https://doi.org/10.3390/ su16229942

Academic Editors: Jesús N. García

Received: 16 October 2024 Revised: 3 November 2024 Accepted: 8 November 2024 Published: 14 November 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). students to virtually tour museums and historical sites [8], VirBELA provides a virtual collaborative environment for learners around the world [9], and platforms such as ClassVR, Engage, and Labster support virtual experiments and interactive learning in multiple disciplines [10–12]. At the same time, technology giants such as NVIDIA and Meta are also promoting sustainable development of the educational metaverse through their technology ecosystems. NVIDIA's Omniverse provides support for precise simulations, and Meta has invested USD 150 million to promote virtual reality education [13-15]. It is estimated that by 2028, the global educational metaverse market will reach USD 43.35 billion, and the user scale is expected to exceed 104.6 million by 2030 [16,17]. In the field of education, more and more scholars are concerned about the ethical impacts and challenges brought about by the integration of artificial intelligence tools into the educational environment [18-20]. In the design industry, more and more designers are beginning to use metaverse technology for virtual creation and team collaboration [21,22]. However, research on the behavioral intentions of design learners in collaborating with AI in the educational metaverse is still relatively limited. Therefore, this study aims to explore the key factors that affect the behavioral intentions of design learners by constructing the concept of a Human-AI collaboration platform in the educational metaverse (EMH-AIc). This will not only help optimize the collaborative experience of design learners and AI but also promote design education toward a more equitable, personalized, and sustainable development direction.

This study proposes that human and artificial intelligence collaboration in the educational metaverse (EMH-AIc) can serve as an effective path for design learners to overcome the limitations of the real world and carry out complex design collaboration. EMH-AIc is defined as a platform which allows design learners to interact and collaborate with AI technology in a virtual educational metaverse, thereby improving the learning experience, collaboration efficiency, and creativity [23–26]. Its specific operating mechanisms include the following. (1) Through high-precision modeling and AI algorithms, design tools and processes are transformed into interactive virtual models to achieve digital twins. (2) Twoway data flow and real-time feedback between the virtual design environment and AI tools are achieved so that learners can obtain intelligent design suggestions. (3) Learners manipulate virtual design projects in real time, and AI technology automatically generates and optimizes design solutions, providing personalized creative experiences and enhancing collaboration efficiency.

UNESCO emphasizes that learners' behavioral intention plays an important role in promoting global education goals [27]. Current research on behavioral intentions in the educational metaverse has mostly focused on basic frameworks such as the technology acceptance model (TAM), focusing on fundamental factors such as usability and ease of use [28]. However, these studies often ignore the key factors at the social and cognitive levels which influence the behavioral intentions of learners in complex virtual learning environments. Therefore, based on social cognitive theory (SCT) in social psychology and education, this study includes the perspective of human-AI collaboration for the first time and proposes a comprehensive behavioral model covering the external environment, individual cognition, and behavioral intention to fill the current research gap in the multidimensional understanding of behavioral intention. SCT believes that individual behavior is the result of the joint action of the external environment and personal cognition. Especially in complex learning situations, individuals need to adjust their behavior through processes such as correctly judging their own abilities, predicting the results of actions, and evaluating social opportunities and constraints [29,30]. This study integrates social and cognitive perspectives to construct a more comprehensive behavioral intention model which not only expands the theoretical framework of behavioral intention but also provides an application reference for design education practice in the educational metaverse and lays a theoretical foundation for sustainable development of the educational metaverse in the field of design education.

This research team aims to identify and analyze the key factors which influence design learners' use of EMH-AIc and provide practical guidance and suggestions for the design and operation of the educational metaverse. By revealing these key factors, this study not only helps the educational metaverse platform attract and retain design learners but also strives to promote lifelong learning and reduce educational inequality. Based on the SCT analysis framework, this study raises three core research questions: (1) What factors influence users' use of EMH-AIc? (2) How do these factors affect users' behavioral intentions? (3) Finally, do these factors promote the development of EMH-AIc? Answering these questions will provide the necessary theoretical and practical support for the continued development of EMH-AIc.

This study is structured as follows. Section 2 discusses the literature review and theoretical background related to EMH-AIc, aiming to lay an academic foundation for the study. Section 3 proposes research hypotheses based on SCT and analyzes the various factors which affect the behavioral intentions of design learners. Section 4 introduces the research methods, including detailed descriptions of the sample selection, questionnaire design, and data collection methods. Section 5 presents the research results, covering the reliability and validity tests of the model and analysis of the hypothesis testing. Section 6 discusses the results of the empirical study and evaluates their impact at the theoretical and practical levels. Section 7 points out the limitations of this study and proposes suggestions for future research directions. Finally, Section 8 summarizes the study. This study strictly followed academic standards to ensure the scientific, validity, and repeatability of the research methods.

2. Theoretical Background and Literature Review

2.1. Human-AI Collaboration in the Educational Metaverse as a Catalyst for Innovation in Design Education

The formation of the educational metaverse stems from the both the drive of technological innovation and the need for educational equity. In 2003, the advent of the Second Life virtual world platform marked the prototype of the educational metaverse. Through an immersive virtual environment, the platform enables users to participate in social, learning, and creative activities in the form of virtual characters and has taken the lead in exploring virtual classrooms, interdisciplinary cooperation, and resource sharing, laying the foundation for future development of the educational metaverse [31,32]. With the rapid development of virtual reality (VR), augmented reality (AR), and artificial intelligence (AI) technologies, the educational metaverse has gradually become an important experimental field for exploring innovative educational models [33].

The introduction of artificial intelligence (AI), especially in personalized learning and real-time interaction, has greatly promoted teaching innovation in the educational metaverse. For example, intelligent learning management systems (LMSs) can use AI algorithms to recommend personalized learning paths for students [34], and natural language processing (NLP) technology supports real-time interaction and question-answering between virtual tutors and students [35,36]. In addition, deep learning technology helps predict students' learning outcomes by analyzing student data and provides teachers with timely strategy adjustment suggestions [37,38]. The application of these technologies not only improves teaching efficiency in the educational metaverse, but also greatly enhances the personalized learning experience [39].

At the same time, AI technology also plays an important role in promoting educational equity, especially in expanding the accessibility of educational resources and narrowing regional educational gaps. For example, online learning platforms such as Coursera and edX use AI technology to provide learners around the world with personalized learning suggestions and automated feedback, enabling students in remote areas to obtain the same quality of educational resources as urban students, thereby effectively narrowing the regional educational gap [40,41].

In design education, the application of AI technology has become a key factor in improving learning outcomes. Through machine learning and deep learning, AI can conduct in-depth analysis of learners' design works and generate personalized design suggestions, thereby helping students optimize the creative process and improve their creative efficiency and work quality [42,43]. For example, Autodesk's Dreamcatcher helps learners explore innovative design solutions and inspire their design inspiration by providing a variety of design options [44]. Similarly, AI design tools such as Adobe Sensei can analyze students' design styles and automatically recommend suggestions, such as color matching and layout adjustments, to help students further improve the expressiveness of their works [45]. These tools not only help students improve their creative efficiency but also enhance their understanding and control of the design process. Studies have shown that AI-assisted learning models can effectively improve learners' participation and learning interest [46]. He et al.'s study introduced AI technology into the traditional teaching system and found that after AI assistance, the creative level and production ability of design students were significantly improved [47]. In addition, the control group experiment of Huang's team further proved that the educational model combining AI technology with the educational metaverse can stimulate learners' enthusiasm more than using AI technology alone. This shows the importance of collaboration between AI and the educational metaverse in promoting innovation in design education.

The synergy between the educational metaverse and AI has brought unprecedented innovation opportunities to design education [48]. AI-generated content (AIGC) has rapidly become one of the key technologies in design education. It can provide personalized creative support based on students' design styles, greatly improving design efficiency. Pan et al. proposed a solution for AI-assisted design work. Students can use AI to achieve intelligent creation in the metaverse, significantly improving the efficiency and creative expression of the design process [49]. Yang et al.'s team verified the impact of the metaverse virtual environment on art design students through a case study method. The results showed that this intelligent collaboration model can significantly improve students' learning outcomes, further proving the potential for wide application of AI and educational metaverse collaboration in design education [50].

Previous studies mostly relied on the technology acceptance model (TAM) or the unified theory of acceptance and use of technology (UTAUT) to evaluate users' acceptance of the educational metaverse, usually focusing on the relationship between perceived usefulness, ease of use, and intention to use [51–53]. For example, Wang et al. explored the intention to use the metaverse platform in the education field based on the TAM model [54], while AlHamad and Akour et al. also conducted similar research on the application of the metaverse platform in higher education, analyzing the impact of technical factors on user intention [55,56]. In addition, Teng et al. extended the UTAUT model to examine learners' willingness to adopt and continue to use the educational metaverse, further explaining the relationship between technology acceptance and behavioral intention [57]. At the same time, the Samed team focused on the behavioral characteristics of Generation Z students in using the educational metaverse, pointing out the unique psychological and behavioral characteristics of this generation in technology acceptance and continued use [58].

Although these studies revealed the importance of technology in promoting development of the educational metaverse, their limitation is that most studies mainly explored users' behavioral intentions from a technical perspective, focusing on individuals' functional perceptions of technology and rarely considering the comprehensive effects of social and cognitive factors on behavior. In the context of AI collaboration in the educational metaverse, learners' behavior is not only affected by technology acceptance but also driven by social influence, self-efficacy, and outcome expectations. Unlike the "acceptance intention" focused on by the TAM and UTAUT, SCT provides a more comprehensive framework which can cover complex factors such as learners' social interaction, individual cognitive regulation, and situational dynamics, which are particularly critical in the context of design education. In particular, in the scenario of AI and design learners collaborating, social cognitive factors are particularly important for understanding learners' continued use intentions and behavioral adjustments, because AI collaboration in design education requires learners to be able to continuously adapt, reflect, and adjust in practice.

5 of 28

Therefore, based on social cognitive theory (SCT), this study aims to fill the gaps in existing research by analyzing key social cognitive factors such as social influence, self-efficacy, and outcome expectations and exploring how these factors jointly influence learners' behavioral intentions in the educational metaverse, where AI and design learners collaborate. The adoption of SCT not only fills the social cognitive gap that the TAM and UTAUT cannot fully explain but also provides a theoretical basis for optimizing the collaboration between AI and learners. This study will lay the foundation for further development of the educational metaverse in design education, making it more in line with learners' social cognitive needs in practice and supporting a more personalized and continuous learning experience.

2.2. Social Cognitive Theory (SCT)

Social cognitive theory (SCT) was proposed by Albert Bandura. It emphasizes that individual behavior, personal cognition, and the external environment interact through "triadic reciprocal determinism" (TRD) to jointly determine individual behavioral choices [59,60]. The core idea of SCT is that individuals gradually form cognition of their own abilities and anticipate the results of future behaviors by observing the behavior of others, social interactions, and situational feedback [61]. Different from traditional passive user behavior theories, SCT provides a dynamic framework which explains how individuals actively make behavioral choices through the interaction of cognition and the external environment [62]. Therefore, based on SCT, this study aims to analyze and design learners' behavioral intentions and promote sustainable development of the educational metaverse, especially in the context of human-AI collaboration.

In recent years, the application of SCT in virtual environments has become increasingly widespread, covering multiple research fields. For example, in the field of healthcare, researchers have constructed an SCT framework which includes environmental dimensions (subjective norms and trust), personal dimensions (motivation and self-efficacy), and behavioral dimensions (behavioral intention) to analyze users' willingness to use remote virtual medical services [63]. In the field of online games, SCT has been used to analyze the impact of factors such as social support and emotional commitment on user loyalty [64], as well as explain users' motivation to watch live games [65]. In the field of education, Stuart's research showed that improving teachers' self-efficacy significantly increased students' intention to use AI technology [66]. In addition, Wijiaya explored the innovative behavior of primary school mathematics teachers and found that promoting conditions and self-efficacy had a direct impact on behavior [67]. Almulla's research evaluated the application of SCT in online education and analyzed the impact of contextual factors on students' learning behavior [68]. In addition, SCT is also widely used to construct teaching strategies to improve teaching effectiveness [69].

Although social cognitive theory (SCT) has been widely used in virtual learning environments, revealing the mechanism by which the external environment (such as technical support and teacher feedback) and personal cognition (such as self-efficacy and outcome expectations) jointly influence user behavior, these studies provide a solid theoretical basis for this study, especially in the analysis of behavioral intentions in designing learners' collaboration with AI. However, in the emerging field of the educational metaverse, systematic research on designing learners' collaboration with AI is still relatively insufficient.

For example, Amany's study explored the intention to use distance education platforms but did not conduct an in-depth analysis of specific educational fields [70]. Abeer focused on the behavioral intention of art students in mobile learning. However, the functions of the mobile learning system in his study were relatively basic, including only support functions such as PowerPoint display, homework, and testing and lacking research on more complex collaborative and interactive functions [71]. Although these studies provide a preliminary background for the application of SCT, there is still a large gap in the systematic analysis of the educational metaverse environment designed for learners to collaborate with AI. Therefore, this study will explore the behavioral intentions of design learners to collaborate with AI in the educational metaverse based on the three-dimensional structure of SCT (external environment, individual cognition, and personal behavior). The research will focus on analyzing the impact of key variables such as social influence, rewards, teacher support, facilitation, self-efficacy, outcome expectations, and trust on the behavioral choices of design learners. By filling the gaps in the existing literature, this study not only aims to optimize the model of collaboration between AI and learners but also provide theoretical support for the development of the educational metaverse and promote the sustainable development of design education.

3. Hypothesis Development

3.1. The Direct Influence of the External Environment on Individual Behavior

The direct influence of the external environment on individual behavior has been validated across multiple theoretical frameworks, including social exchange theory (SET), social cognitive theory (SCT), and the theory of planned behavior (TPB) [72–74]. These frameworks collectively emphasize that behavior is shaped not only by internal cognitive factors but also by strong external environmental influences. Previous studies have confirmed that the external environment plays a crucial role in shaping the behavioral intentions of both teachers and students [75,76]. Given this understanding, it is essential to explore how external factors directly influence the behavioral intentions of design learners in virtual learning environments, particularly in the context of AI collaboration. In this study, the external environment consists of four major elements: social influence, rewards, teacher support, and facilitating conditions. Social influence is defined as the impact of others' opinions or behaviors (such as peers, teachers, or social groups) on learners, including encouragement, feedback, and recognition. Research by Salloum et al. and Lakhal et al. confirmed that social influence positively affects the intention to use e-learning systems within the UTAUT framework [77,78]. Rewards refer to material or non-material incentives, such as prizes, badges, or points. According to social exchange theory, individuals tend to make behavioral choices based on potential rewards or benefits [79]. Saranya et al. demonstrated that user behaviors, such as writing online reviews, increased significantly when driven by rewards [80]. Teacher support is another critical factor, defined as guidance and feedback provided by teachers to help learners collaborate with AI on design tasks within the educational metaverse. Granziera et al.'s cross-national research found that teacher support significantly boosts students' academic motivation and behavioral intentions [81]. Sakiz's findings further support the positive impact of teacher support on behavioral intention [82]. Finally, facilitating conditions refer to the availability of resources and technological support for learners to engage with AI within the educational metaverse. Nikou et al. demonstrated that the availability of devices and platform usability greatly affect learners' usage intentions in mobile learning systems [83]. Based on these insights, this study empirically investigates how these external environmental factors impact design learners' behavioral intentions to use the EMH-AIc platform and proposes the following hypotheses:

- **H1.** Social influence positively affects the behavioral intention of design learners to use EMH-AIc.
- **H2.** Rewards positively affect the behavioral intention of design learners to use EMH-AIc.
- **H3.** Teacher support positively affects the behavioral intention of design learners to use EMH-AIc.

H4. Facilitating conditions positively affect the behavioral intention of design learners to use EMH-AIc.

3.2. The Influence of the External Environment on Individual Cognition

After being influenced by the external environment, learners usually make psychological and cognitive assessments to determine whether the environment is conducive to their learning and development [84,85]. The individual cognitive dimensions in this study include three main factors: self-efficacy, outcome expectations, and trust. Self-efficacy in this study refers to the confidence of design learners in their ability to successfully collaborate with AI technology to complete design tasks in the virtual educational metaverse. Outcome expectations are defined as the expectations of design learners for the results which can be obtained by using the AI collaboration platform, including whether it can improve design capabilities and learning outcomes. Trust refers to learners' trust in the educational metaverse platform, especially in terms of collaboration support, privacy protection, and data security. These factors represent the confidence of design learners in collaborating with AI technology to complete design tasks in the virtual educational metaverse (EMH-AIc), their expectations for the expected results of the collaboration platform, and their trust in the platform.

Social influence plays an important role in improving learners' self-efficacy, outcome expectations, and trust [63,86]. According to Bandura's social cognitive theory, positive feedback and support from others can significantly enhance an individual's self-efficacy [87]. In the educational metaverse, when learners observe their peers and mentors' positive comments on and frequent use of the platform, their self-confidence will increase, and they will be more inclined to believe they can successfully use the platform. In addition, motivation theory points out that learners' behaviors and expectations are often influenced by the feedback of others. When they observe that others have achieved significant learning results through the platform, their outcome expectations will also increase [85]. Trust, as the basis for users to continue to use the platform, will be enhanced by positive demonstrations of social influence [88]. When learners see that their peers and mentors have a high degree of recognition of the platform, they will have more confidence in the security and reliability of the platform. Based on this, this study proposes the following hypothesis:

H5. Social influence has a positive impact on design learners' self-efficacy (H5a), outcome expectations (H5b), and trust (H5c) in using EMH-AIc.

As an external motivational factor, rewards have a significant impact on learners' selfefficacy, outcome expectations, and trust. Self-determination theory shows that rewards can not only stimulate learners' intrinsic motivation but also convey affirmation of their abilities, thereby enhancing their self-efficacy [89]. At the same time, the reward mechanism increases learners' attention to the platform, making them pay more attention to the potential benefits brought by the platform, thereby improving their outcome expectations [90]. In addition, rewards can also serve as the platform's recognition of users' positive behaviors and enhance users' trust. Studies have shown that when learners receive reward feedback from the platform, they are more likely to believe that the platform recognizes the value of their behavior, thereby enhancing their trust in the platform [91,92]. Based on this, this study proposes the following hypothesis:

H6. *Rewards have a positive impact on design learners' self-efficacy (H6a), outcome expectations (H6b), and trust (H6c) in using EMH-AIc.*

Teacher support can effectively enhance learners' self-efficacy, outcome expectations, and trust by providing professional guidance and emotional support. As knowledge authorities, teachers can help learners overcome their fear of AI technology, improve their self-confidence, and make them believe that they can successfully use the platform [93,94]. In addition, teachers can help students understand the advantages of the platform more clearly through demonstration and guidance, thereby improving their outcome expectations. Teachers' recommendations and recognition also tend to enhance students' trust in the

platform, especially in terms of data security and privacy protection [95,96]. Based on this, this study proposes the following hypothesis:

H7. *Teacher support has a positive impact on design learners' self-efficacy (H7a), outcome expectations (H7b), and trust (H7c) in using EMH-AIC.*

Convenience can provide technical support and resource guarantees, which can significantly improve their self-efficacy, outcome expectations, and trust [97]. Convenience helps learners overcome technical barriers by reducing the difficulty of using the platform, making them more confident in their ability to use it. Convenient usage conditions not only reduce the operational burden of learners but also make them more expectant of the platform's effects. In addition, the provision of convenience enhances the reliability of the platform, thereby enhancing learners' trust in the platform and making them more willing to use the platform in the long term [98]. Based on this, this study proposes the following hypothesis:

H8. *Facilitating conditions have a positive impact on design learners' self-efficacy (H8a), outcome expectations (H8b), and trust (H8c) in using EMH-AIc.*

3.3. The Direct Impact of Individual Cognition on Individual Behavior

Based on the theory of reasoned action (TRA), individual behavior is shaped by attitudes toward the behavior, beliefs regarding the behavioral outcomes, and evaluations of those outcomes [99]. Within this framework, learners' cognitive assessments play a pivotal role in forming their behavioral intentions. As learners are influenced by external factors, such as social influence, rewards, teacher support, and facilitating conditions, they integrate these external inputs with internal cognitive factors like self-efficacy, outcome expectations, and trust to form a clear intention to engage in specific behaviors. In this study, behavioral intention refers to whether design learners, after being exposed to these external factors and engaging in cognitive evaluations, are inclined to collaborate with AI in the educational metaverse to complete design tasks. Previous research highlights that selfefficacy, outcome expectations, and trust are key cognitive factors influencing behavioral intention. For instance, high self-efficacy is linked to stronger learning motivation and a greater willingness to tackle complex tasks within virtual learning environments [63]. Similarly, when learners hold high expectations about the outcomes of using a specific technology, such as believing it will enhance their learning or design skills, their intention to engage with the platform significantly increases [100]. Trust is also critical; learners who have confidence in the reliability and security of a virtual platform, including its AI functionalities, are more likely to engage deeply with the system [101]. Based on this, this study will empirically test the direct impact of the three cognitive factors of self-efficacy, outcome expectations, and trust on design learners' behavioral intentions to use EMH-AIc. Based on the above analysis, the following hypotheses are proposed:

H9. Self-efficacy positively influences design learners' behavioral intentions to use EMH-AIc.

H10. Outcome expectations positively influence design learners' behavioral intentions to use EMH-Alc.

H11. *Trust positively influences design learners' behavioral intentions to use EMH-AIc.*

3.4. Proposed Research Model

This study developed a multidimensional conceptual model based on social cognitive theory (SCT), integrating social exchange theory and planned behavior theory to explore the impact of external environment factors (such as social influence, rewards, teacher support, and facilitating conditions) on cognitive factors and behavioral intention. The model aims

to provide a theoretical basis for optimizing the educational metaverse platform, promoting educational equity, and enhancing learners' AI collaboration experiences.

4. Research Methods

4.1. Questionnaire Development

The research model is composed of eight key variables: social influence (SIE), rewards (RW), teacher support (TS), facilitating conditions (FC), self-efficacy (SE), outcome expectations (OE), trust (TRU), and behavioral intention (BI). Each of these variables has been adapted from previously validated scales in the literature to ensure their relevance in the context of the educational metaverse for human-AI collaboration (EMH-AIc). SIE items were adapted by Nikou et al.'s study [83,102,103]. RW items were modified based on Cui et al.'s study [80,104,105]. TS items were derived from Adela et al.'s study [106–109]. FC items were adapted according to Ain et al.'s study [83,97,110]. SE items were modified following Shahangian et al.'s study [63,75,86,103,111]. OE items were adapted based on the research by Rana et al. [86,100,103,111]. TRU items were adapted from Wu et al.'s study [63,112]. BI items were based on Li et al.'s study [63,83,102,103,110]. Details of the survey items related to these variables are provided in Appendix A. Each item in the questionnaire was measured using a 5 point Likert scale, ranging from "strongly disagree (1)" to "strongly agree (5)".

Before initiating the survey, three professional translators with extensive experience reviewed the questionnaire to ensure that the content was clear and free from translation errors which could cause misunderstandings. Additionally, three professors specializing in education and two industry experts examined the content of the questionnaire to ensure its reliability and validity.

The questionnaire aims to evaluate participants' initial reactions and usage intentions after experiencing the EMH-AIc platform, helping us assess their likelihood of adopting such platforms [113]. Selecting the appropriate platform was critical, and after careful consideration of EMH-AIc's unique characteristics, we chose the Apache Art Studio on Spatial.io as the experimental platform (Figure 1). This decision was driven by several key factors:



Figure 1. Apache Art Studio user experience and practical scenarios.

- Accessibility and User Engagement: Spatial is browser-based, requiring no extensive setup or VR hardware. This feature is particularly valuable in educational research, as it allows participants from diverse backgrounds, including art professionals and students, to easily access the platform, thus lowering the technical barrier.
- Real-time Collaboration: Spatial supports interactive real-time collaboration, a key element in studying human-AI collaboration. Users can upload digital artwork, interact with 3D objects, and co-create in shared spaces, making it an ideal environment for exploring collaborative processes between human users and AI-driven tools.

- Creative and Artistic Applications: Apache Art Studio emphasizes art and creative exploration, aligning with the needs of design learners and professionals. The platform allows users to showcase, manipulate, and engage with artistic content, providing a practical environment to study how AI can assist or enhance the creative process in educational and artistic contexts.
- Integration of AI Tools: While Spatial focuses on user-generated content, it has the potential to integrate AI-driven content creation and curation tools. This makes it an excellent environment for studying how AI can assist or enhance human creativity in educational or professional artistic settings. For instance, users can explore how AI helps curate virtual exhibitions, generate artistic assets, or facilitate cross-cultural collaboration among artists.
- Multimodal Learning Environment: Spatial provides an immersive, multimodal environment combining audio, visual, and interactive elements. This set-up supports the integration of AI technologies into a rich learning ecosystem, encouraging not only artistic exploration but also educational activities like workshops or collaborative design sessions.

4.2. Data Collection and Analysis

This study aims to explore design learners' cognitive evaluations when using the educational metaverse for human-AI collaboration (EMH-AIc) and how these evaluations influence their behavioral intentions. The focus is on how four external factors—social influence, rewards, teacher support, and facilitating conditions—drive design learners' willingness to use the EMH-AIc platform by influencing their self-efficacy, outcome expectations, and trust. To validate the research model, an online anonymous survey was conducted using snowball sampling [114].

Snowball sampling is an efficient way to reach a specific and concentrated target group. It is particularly suitable for the subjects of this study, namely design learners, especially those who are familiar with the metaverse and AI collaboration tools [115]. This group accounts for a small proportion of the total population and is concentrated in specific design schools. Therefore, we launched the first round of questionnaire invitations among design students from three top design schools in China and spread them through their social networks to obtain effective samples which met the research needs. This method not only improved the efficiency and pertinence of data collection but also ensured that the sample was highly representative of the target group. However, the limitation of snowball sampling is that it may introduce sample homogeneity bias and selection bias. Since the respondents mainly came from schools and majors with similar backgrounds, the behaviors and attitudes of the respondents may have been relatively consistent, thus limiting the diversity of the sample. In addition, expanding the sample through social networks may have led to the exclusion of some individuals who had not been exposed to the network, further affecting the representativeness of the sample. In order to minimize the impact of these biases on the research conclusions, we strictly explained the group characteristics of the sample in the data analysis and discussion and made it clear that it had a certain degree of concentration characteristics to avoid over-generalization of the research conclusions. In addition, the individual characteristics of the sample were recorded in detail and considered in the analysis to increase the broad applicability and rigor of the findings. Although snowball sampling has its limitations, we took scientific control measures to ensure the credibility and explanatory power of the conclusions of this study in the design education context.

The survey was divided into two parts. The first part required participants to experience the EMH-AIc platform, ensuring they were familiar with its features and functionality before answering the questionnaire. For this, each participant was provided with a detailed user guide and access link, directing them to the virtual space within Apache Art Studio on the Spatial.io platform. There, participants engaged in design creation and collaboration in a metaverse environment. This virtual space allowed users to explore with digital avatars, participate in interactive discussions via video, voice, or text, upload design projects, and collaborate with AI tools. Each participant was asked to spend a minimum of 15 min on the platform to ensure sufficient depth and breadth of the experience. The platform was accessible on Windows, macOS, iOS, and Android systems, allowing participants to collaborate in real time using virtual tools. After completing the experience in Apache Art Studio, the participants proceeded to the second part of the questionnaire. To ensure data accuracy, all participants were required to complete both the EMH-AIc experience and the questionnaire on the same day, minimizing any potential impact of memory recall on the accuracy of their responses.

A total of 550 questionnaires were collected for this study. After reviewing the responses, 17 were excluded due to uniform or invalid answers, leaving 533 valid samples for data analysis. Table 1 provides the demographic characteristics of the respondents. Males accounted for 58.30% (311 individuals) of the respondents, while females made up 41.70% (222 individuals). In terms of age distribution, the majority of the participants fell within the 18-29 age group (74.11%) followed by those aged 30-39 (21.01%), with fewer respondents aged 40 and above (4.88%). This distribution reflects the typical age structure of the design learner group, among which those aged 18–29 are in the early stages of university study and career development. Learners at this stage are usually more receptive to emerging technologies and more actively seek to combine technology with learning, especially in the field of design. The proportion of learners between 30 and 39 was also considerable, indicating that some learners still chose to continue learning or improve their design skills through new technical means after working for several years. This may also reflect their need for career advancement and skill updating. The proportion of learners aged 40 and above was relatively low, which may be related to the fact that design practitioners in this age group have been relatively stable in their career development, or their interest in and acceptance of new technologies are relatively low. Therefore, the distribution of this survey result is consistent with the learning motivation and behavioral characteristics of different age groups in the process of design education and career development.

Measure	Items	Frequency	Gender (Male)	Gender (Female)	Percentage (%)
	18~29	395	230	165	74.11
Age	30-39	112	66	46	21.01
0	40 and above	26	15	11	4.88
	Less than undergraduate	60	36	24	11.30
E1 1	Undergraduate	329	189	140	61.70
Education ¹	Post-graduate	104	59	45	19.50
	Doctor	40	27	13	7.50
	Graphic design or visual communication design	60	34	26	11.30
	Industrial design or product design	105	63	42	19.70
	Interaction design user experience design	60	32	28	11.30
	Environmental design	70	45	25	13.10
Major ²	Fashion design	59	36	23	11.10
,	Digital media design	93	50	43	17.40
	Service design	31	18	13	5.80
	Social innovation design	13	8	5	2.40
	Emerging technology and design	42	25	17	7.90
	1–2 times per month	49	25	24	9.20
T	3–4 times per month	103	50	53	19.30
Usage	1–2 times per week	200	122	78	37.50
frequency	3–5 times per week	123	78	45	23.10
	Daily	58	36	22	10.90
	Only used in class	82	42	40	15.40
C · · ·	Occasionally used outside class	235	141	94	44.10
Context	Frequently used for both class and non-class activities	165	102	63	30.90
aependence °	Used every day for learning or collaboration	51	26	25	9.60
	Total participants	533	311	222	100.0

Table 1. Demographics of participants (N = 533).

¹ The measure of education includes graduated and currently enrolled. ² Choose similar option if major names differ. ³ Motivation and scenarios.

In terms of educational background, the majority of the respondents were undergraduate students (61.70%), followed by graduate students (19.50%), and doctoral students (7.50%). This indicates that most respondents were in the early stages of their educational or career development, with a significant portion pursuing advanced degrees, which may explain their greater exposure to educational technologies like the metaverse. It is worth noting that there were relatively more male undergraduates, which may be related to the fact that some technology-oriented design majors (such as industrial design) have more male participants. Regarding professional focus, participants specializing in industrial design or product design (19.70%) and digital media design (17.40%) were represented the most. This suggests that these fields have a higher demand for technological integration, particularly in AI collaboration during the design process, making these learners more inclined to use platforms like EMH-AIc. As for platform usage frequency, 37.50% of the participants reported using the platform 1–2 times per week, while 23.10% used it 3–5 times per week. Although a significant portion of design learners use the platform regularly, it has not yet become a daily tool for most. Additionally, 44.10% of the respondents stated they occasionally used it outside of class, while 31% used it frequently for both classroom and extracurricular activities. This reflects the flexibility and wide applicability of the educational metaverse platform in both formal learning environments and self-directed collaboration.

This study used partial least squares structural equation modeling (PLS-SEM) to evaluate the research model and verified the validity of the measurement model and structural model through SPSS 27.0 and SmartPLS 4.0. PLS-SEM is gradually gaining attention in the study of user experience and behavioral intention in the context of the metaverse due to its advantages in predicting dependent variables and explaining the relationship between independent variables [116–118]. In addition, since PLS shows strong robustness and accuracy in the case of small sample data, it is particularly suitable for the data collected in this study [119].

5. Results

All variables in this study were collected through the same questionnaire, and thus the common method variance (CMV) was tested in the first step of statistical analysis to eliminate potential bias problems. Harman's one-factor test was used to evaluate the CMV, and factor analysis was performed on all items in the model using IBM SPSS Statistics 27.0. A total of eight factors were extracted in this study, and the first factor explained 28.58% of the variance, which was less than 50%. Therefore, the data in this study did not have significant common method variance problems [120].

5.1. Measurement Model

To evaluate the model fit in this study, we used the standardized root mean square residual (SRMR) and the normed fit index (NFI) as the primary indicators (Table 2). Model fit determines the accuracy and effectiveness of the model in explaining the data, making these metrics crucial in assessing model quality. The SRMR value was 0.047, indicating a good model fit. Typically, an SRMR value below 0.08 is considered an acceptable fit, and values under 0.05 often suggest an excellent fit between the model and actual data [119]. Therefore, this study's model demonstrated a strong fit with minimal data error, suggesting that it reasonably explained the relationships between the variables. The NFI, used to compare model fit with a null model, approaches one when the fit is good [121]. In this study, the NFI value was 0.866, further indicating a relatively strong fit for the model.

Table	2.	Model	fit.
-------	----	-------	------

Fit Index	Computed Values	Threshold Reference
SRMR	0.047	[119]
NFI	0.866	[121]

In the partial least squares structural equation modeling (PLS-SEM) analysis, we primarily examined the reliability, internal consistency reliability, convergent validity, and discriminant validity of the measurement indicators. Factor loadings indicate how well a measurement variable reflects the information of the latent variable, with higher values (closer to one) suggesting stronger representation. As shown in Appendix A and Table 3, all factor loadings for the measurement variables were above 0.7, with the factor loadings for the structural elements ranging from 0.788 to 0.88. These values confirm that the measurement model demonstrated a strong representation of the latent constructs, ensuring both the reliability and validity of the model.

Table 5. Factor loads and cross-loads.	Table 3. Factor loads and	cross-loads.
--	---------------------------	--------------

	BI	FC	OE	RW	SE	SIE	TRU	TS
BI1	0.808	0.262	0.244	0.33	0.356	0.225	0.375	0.276
BI2	0.847	0.285	0.321	0.337	0.387	0.264	0.35	0.255
BI3	0.833	0.354	0.257	0.272	0.378	0.269	0.366	0.288
BI4	0.829	0.275	0.256	0.297	0.316	0.251	0.391	0.275
FC1	0.275	0.845	0.181	0.249	0.322	0.275	0.339	0.241
FC2	0.312	0.788	0.158	0.263	0.294	0.187	0.318	0.267
FC3	0.296	0.826	0.182	0.215	0.297	0.202	0.311	0.262
FC4	0.287	0.832	0.205	0.249	0.289	0.252	0.324	0.298
OE1	0.29	0.195	0.856	0.233	0.261	0.207	0.282	0.181
OE2	0.254	0.19	0.834	0.311	0.243	0.246	0.271	0.194
OE3	0.305	0.19	0.847	0.317	0.292	0.241	0.306	0.23
OE4	0.239	0.163	0.825	0.231	0.204	0.211	0.299	0.185
RW1	0.317	0.239	0.258	0.83	0.295	0.233	0.328	0.23
RW2	0.331	0.269	0.307	0.832	0.291	0.26	0.357	0.245
RW3	0.296	0.216	0.256	0.837	0.272	0.238	0.343	0.225
RW4	0.295	0.261	0.268	0.83	0.285	0.235	0.374	0.279
SE1	0.366	0.316	0.26	0.328	0.838	0.245	0.269	0.291
SE2	0.357	0.298	0.246	0.25	0.795	0.26	0.266	0.31
SE3	0.31	0.282	0.239	0.265	0.802	0.256	0.256	0.294
SE4	0.37	0.276	0.216	0.268	0.814	0.238	0.251	0.304
SE5	0.361	0.313	0.262	0.287	0.823	0.238	0.245	0.279
SIE1	0.282	0.24	0.249	0.245	0.235	0.812	0.263	0.198
SIE2	0.18	0.23	0.167	0.21	0.234	0.809	0.295	0.236
SIE3	0.231	0.206	0.224	0.213	0.236	0.819	0.253	0.208
SIE4	0.287	0.231	0.232	0.272	0.28	0.816	0.26	0.204
TRU1	0.356	0.295	0.263	0.327	0.253	0.272	0.798	0.225
TRU2	0.382	0.327	0.294	0.359	0.269	0.307	0.844	0.234
TRU3	0.358	0.336	0.308	0.352	0.256	0.276	0.84	0.243
TRU4	0.387	0.346	0.28	0.362	0.273	0.237	0.843	0.287
TS1	0.304	0.294	0.207	0.257	0.316	0.245	0.252	0.856
TS2	0.295	0.297	0.201	0.299	0.344	0.214	0.276	0.88
TS3	0.246	0.24	0.202	0.194	0.269	0.206	0.236	0.838

In this study, reliability and validity assessments of the variables were conducted using Cronbach's alpha and the composite reliability (CR) and average variance extracted (AVE) as key indicators (Table 4). Cronbach's alpha was employed to measure the internal consistency reliability of each variable. All variables showed a Cronbach's alpha value above 0.80, indicating strong internal consistency, with values ranging from 0.822 (TS) to 0.873 (SE). Composite reliability (CR) was also used to assess the reliability of the latent variables, as it provides a more accurate measure of consistency than Cronbach's alpha. All CR values exceeded 0.85, indicating high reliability, with CR values ranging from 0.887 (SIE) to 0.908 (SE). Finally, AVE was used to measure the convergent validity of each variable, with all AVE values above 0.50, indicating that each variable explained more than half of the variance in its indicators and thus meeting the standard for convergent validity [122]. The AVE values ranged from 0.662 (SIE) to 0.737 (TS), demonstrating good discriminant and convergent validity for each variable. Consequently, the reliability and validity assessments in this study meet the accepted standards, ensuring the robustness of the measurement model.

	CA	CR (rho_a)	CR (rho_c)	AVE
BI	0.849	0.849	0.898	0.688
FC	0.841	0.841	0.894	0.677
OE	0.862	0.867	0.906	0.707
RW	0.852	0.853	0.9	0.692
SE	0.873	0.874	0.908	0.663
SIE	0.83	0.833	0.887	0.662
TRU	0.851	0.852	0.9	0.691
TS	0.822	0.827	0.893	0.737

Table 4. Indicators of the reliability and validity of the concept.

Based on the discriminant validity analysis in Table 5, we compared the square roots of the AVE for each latent variable with the correlation coefficients between the latent variables. The bolded values along the diagonal represent the square root of the AVE for each latent variable, while the off-diagonal values are the correlation coefficients between the latent variables. To demonstrate discriminant validity in the model, the square root of the AVE for a given latent variable should be greater than its correlation coefficients with other latent variables. As can be seen in the table, all latent variables met this criterion. For instance, the square root of the AVE for behavioral intention (BI) was 0.829, higher than its correlation coefficients with other latent variables such as facilitating conditions (FC), outcome expectations (OE), and rewards (RW). Similarly, the square root of the AVE for self-efficacy (SE) was 0.814, exceeding its correlations with other variables. This indicates that each latent variable in the model possessed strong discriminant validity, meaning that each variable could be effectively distinguished from the others and had independent explanatory power. Therefore, from the perspective of discriminant validity, the measurement model in this study demonstrated a high capacity for differentiating between variables, meeting the requirements for structural equation modeling. This enhanced the reliability and validity of the model's results.

	BI	FC	OE	RW	SE	SIE	TRU	TS
BI	0.829							
FC	0.355	0.823						
OE	0.326	0.221	0.841					
RW	0.373	0.297	0.328	0.832				
SE	0.434	0.365	0.300	0.344	0.814			
SIE	0.305	0.279	0.27	0.291	0.304	0.814		
TRU	0.446	0.393	0.344	0.421	0.316	0.328	0.831	
TS	0.330	0.325	0.237	0.295	0.363	0.259	0.298	0.858

Table 5. Distinguishing validity (Fornell–Larcker criterion).

In order to further ensure the discriminant validity of the model, we supplemented the Heterotrait:monotrait (HTMT) ratio analysis in addition to the Fornell–Larcker standard. The HTMT ratio is a more stringent discriminant validity assessment method which judges the discrimination between variables by comparing the average values of the heterogeneity and homogeneity relationships between latent variables. Generally, HTMT values below 0.85 or 0.90 indicate good discriminant validity [123,124].

In this study, we used the HTMT ratio to verify the independence between the latent variables (Table 6). The results showed that the HTMT values of all variable pairs were lower than 0.85, indicating that under stricter standards, the latent variables still had significant discriminant validity. Therefore, through the dual verification of the Fornell–Larcker standard and the HTMT ratio, the discriminant validity of the measurement model of this study was further supported, which enhanced the independence between the latent variables and the reliability and validity of the model results.

	BI	FC	OE	RW	SE	SIE	TRU	TS	
BI	-								
FC	0.42								
OE	0.378	0.258							
RW	0.438	0.349	0.377						
SE	0.503	0.426	0.343	0.398					
SIE	0.358	0.332	0.315	0.343	0.355				
TRU	0.526	0.464	0.402	0.494	0.367	0.392			
TS	0.393	0.388	0.28	0.348	0.426	0.314	0.355	-	

Table 6. Distinguishing validity (HTMT values).

Table 7 shows the R^2 , adjusted R^2 , and Q^2 values of each latent variable in the model. The R^2 value indicates the explanatory power of the independent variable on the latent variable. Among them, BI (0.341), TRU (0.29), and SE (0.255) all reached a moderate level, indicating that the independent variables had a certain explanatory power for these latent variables. Although the R^2 value of OE was relatively low (0.159), it was still reasonable for exploratory research. The adjusted R^2 value was close to the R^2 value, indicating that the model had no obvious noise effect and had good explanatory stability. In addition, the Q^2 values of all latent variables were greater than zero, (For example, BI was 0.228, and TRU was 0.195) indicating that the model had effective predictive relevance. These results show that although the individual values were not high, the model as a whole had moderate explanatory power and predictive ability, supporting its applicability in measurement and structural models.

Table 7. Values of R2 and Q2.

	R ²	R ² Adjusted	Q ²
BI	0.341	0.333	0.228
OE	0.159	0.152	0.107
SE	0.255	0.249	0.166
TRU	0.29	0.284	0.195

5.2. Modeling Analysis

The study's final model demonstrated robust reliability and validity, with variance inflation factor (VIF) values ranging between 1.626 and 2.206 across all indicators, indicating no multicollinearity issues [123–125]. This reliability in our constructs supports the validity of our path analyses, and it enhanced the interpretability of the findings. Our model testing, through bootstrapping (5000 iterations), confirmed the majority of the hypothesized relationships, validating the model's effectiveness in assessing factors influencing behavioral intentions in the EMH-AIc platform.

In the path relationship analysis of variables in Table 8 and Figure 2, H1 was not supported ($\beta = 0.063$, T value = 1.37, p > 0.05), indicating that social influence had no significant direct impact on behavioral intention. In contrast, suppose H2 ($\beta = 0.104$, T value = 2.334, p < 0.05), H3 ($\beta = 0.087$, T value = 2.012, p < 0.05) and H4 ($\beta = 0.093$, T value = 2.131, p < 0.05) were supported, indicating that rewards, teacher support, and convenient conditions have a significant positive impact on behavioral intentions. In addition, social influence had a significant positive impact on self-efficacy, outcome expectations, and trust, supporting hypotheses H5a ($\beta = 0.140$, T value = 3.036, p < 0.05), H5b ($\beta = 0.156$, T value = 3.335, p < 0.05), and H5c ($\beta = 0.156$, T value = 3.668, p < 0.05).

Hypothesis	β	SD	T Value	p Value	Results
H1: SIE -> BI	0.063	0.046	1.37	0.171	Not supported
H2: RW -> BI	0.104	0.044	2.334	0.02	Supported
H3: TS -> BI	0.087	0.043	2.012	0.044	Supported
H4: FC -> BI	0.093	0.043	2.131	0.033	Supported
H5a: SIE -> SE	0.140	0.046	3.036	0.002	Supported
H5b: SIE -> OE	0.156	0.047	3.335	0.001	Supported
H5c: SIE -> TRU	0.156	0.043	3.668	0	Supported
H6a: RW -> SE	0.181	0.046	3.921	0	Supported
H6b: RW -> OE	0.230	0.047	4.941	0	Supported
H6c: RW -> TRU	0.277	0.044	6.357	0	Supported
H7a: TS -> SE	0.206	0.048	4.321	0	Supported
H7b: TS -> OE	0.104	0.046	2.261	0.024	Supported
H7c: TS -> TRU	0.100	0.044	2.264	0.024	Supported
H8a: FC -> SE	0.206	0.048	4.277	0	Supported
H8b: FC -> OE	0.075	0.045	1.656	0.098	Not supported
H8c: FC -> TRU	0.235	0.046	5.143	0	Supported
H9: SE -> BI	0.216	0.049	4.401	0	Supported
H10: OE -> BI	0.093	0.045	2.059	0.04	Supported
H11: TRU -> BI	0.219	0.057	3.833	0	Supported

Table 8. Hypothesis testing and collinearity assessment results.



Figure 2. Results of PLS structural model.

Similarly, the effects of rewards on self-efficacy, outcome expectations, and trust were all significant, supporting hypotheses H6a (β = 0.181, T value = 3.921, *p* < 0.05), H6b (β = 0.230, T value = 4.941, *p* < 0.05), and H6c (β = 0.277, T value = 6.357, *p* < 0.05). The significant impact of teacher support on self-efficacy, outcome expectations, and trust support hypotheses H7a (β = 0.206, T value = 4.321, *p* < 0.05), H7b (β = 0.104, T value = 2.261,

p < 0.05), and H7c ($\beta = 0.1$, T value = 2.264, p < 0.05). Facilitating conditions had a significant impact on self-efficacy and trust (H8a supported, with $\beta = 0.206$ and T value = 4.277, p < 0.05; H8c supported, with $\beta = 0.235$ and T value = 5.143, p < 0.05), but the outcome expectations from the effect were not significant (H8b was not supported, with $\beta = 0.075$ and T value = 1.656, p > 0.05).

Finally, cognitive factors had a significant direct impact on behavioral intention, including self-efficacy (H9, β = 0.216, T value = 4.401, *p* < 0.05), outcome expectations (H10, β = 0.093, T value = 2.059, *p* < 0.05), and trust (H11, β = 0.219, T value = 3.833, *p* < 0.05), with all supporting the positive effect on behavioral intention.

5.3. Mediation Analysis

According to the analysis of the mediation effect results (Table 9), some external factors had a significant indirect effect on behavioral intention (BI) through self-efficacy (SE) and trust (TRU). The mediating effects of social influence (SIE) on BI through SE and TRU were 21.13% and 23.94%, respectively, indicating that when learners receive support from the social environment, this support can effectively enhance their self-efficacy and trust in the platform, thereby indirectly increasing their willingness to use the platform. Similarly, the mediating effect of rewards (RW) on BI through TRU reached 27.11%, indicating that an appropriate reward mechanism can further promote learners' acceptance and intention to use the platform by enhancing their sense of trust. These results emphasize the role of trust as a key mediating variable and verify the importance of trust in a complex technological environment.

Table 9. Mediation effect analysis: path coefficients, significance, and VAF values.

Relationship	β	SD	T Value	p Value	2.50%	97.5%	Results	VAF
SIE -> SE -> BI	0.03	0.012	2.468	0.014	0.009	0.058	Significant Mediation	21.13%
SIE -> OE -> BI	0.014	0.008	1.77	0.077	0.001	0.033	Non-Significant Mediation	9.86%
SIE -> TRU -> BI	0.034	0.013	2.57	0.01	0.013	0.065	Significant Mediation	23.94%
RW -> SE -> BI	0.039	0.014	2.879	0.004	0.016	0.069	Non-Significant Mediation	17.33%
RW -> OE -> BI	0.021	0.011	1.877	0.061	0.001	0.046	Non-Significant Mediation	9.33%
RW -> TRU -> BI	0.061	0.019	3.275	0.001	0.028	0.099	Significant Mediation	27.11%
TS -> SE -> BI	0.045	0.014	3.115	0.002	0.02	0.075	Significant Mediation	27.61%
TS -> OE -> BI	0.01	0.007	1.357	0.175	0	0.027	Non-Significant Mediation	6.13%
TS -> TRU -> BI	0.022	0.011	1.921	0.055	0.003	0.048	Non-Significant Mediation	13.5%
FC -> SE -> BI	0.044	0.014	3.142	0.002	0.02	0.075	Significant Mediation	22.45%
FC -> OE -> BI	0.007	0.006	1.241	0.215	-0.001	0.02	Non-Significant Mediation	3.57%
FC -> TRU -> BI	0.051	0.016	3.158	0.002	0.023	0.087	Significant Mediation	26.02%

In addition, the indirect effect of teacher support (TS) on BI through SE reached 27.61%, indicating that the guidance and support of teachers played a positive mediating role in enhancing learners' self-confidence and willingness to use the platform. On the other hand, the mediating effect of facilitating conditions (FC) on BI through SE and TRU was 22.45% and 26.02%, respectively, indicating that the convenience of technology and resources can significantly improve learners' self-efficacy and trust, thereby indirectly improving their intention to use the platform. However, the mediating effects of other paths such as SIE -> OE -> BI, RW -> OE -> BI, and TS -> OE -> BI did not reach a significant level, indicating that the mediating role of outcome expectations (OE) in these paths was relatively weak. These findings overall highlight the core role of self-efficacy and trust as important mediating variables and provide empirical evidence to support the optimal design of educational platforms.

5.4. Research Models

Figure 3 presents the results of the 19 path coefficients and their significance levels. Behavioral intention in using the educational metaverse was significantly influenced by



Figure 3. Results of research model (*** *p* < 0.001; ** *p* < 0.01; * *p* < 0.05; n.s. = not significant).

6. Discussion

Based on social cognitive theory (SCT), this study explored the key factors which affect design learners' use of a human-AI collaboration platform (EMH-AIc) in the educational metaverse. By constructing a triadic reciprocity framework, the three dimensions of the external environment, cognitive factors, and behavioral intention were integrated to reveal the mechanism which affects the formation of behavioral intention. The model contains eight variables: social influence (SIE), rewards (RW), teacher support (TS), facilitating conditions (FC), self-efficacy (SE), outcome expectations (OE), trust (TRU), and behavioral intention (BI). Although two hypotheses were not supported (i.e., H1 and H8b), the overall research results provide theoretical and empirical support for understanding how design learners form platform usage intentions through external environment and personal cognitive factors and have guiding significance for practical applications.

First, the results reveal the practical implications of external factors on behavioral intention. Social influence did not directly or significantly affect behavioral intention, which suggests that in actual educational environments, design learners' willingness to use the platform depends more on intrinsic motivation and personalized needs rather than social pressure. This conclusion is consistent with the research results of Dahri and Arain et al. [126,127] but different from the conclusions of Altameemi and Wiangkham et al. on social influence [117,128]. This suggests that when promoting new technologies, educational institutions should pay more attention to stimulating learners' interests and intrinsic motivation rather than relying solely on peer or external pressure to promote use. However, social influence indirectly had a significant effect on behavioral intention by enhancing cognitive factors such as self-efficacy and trust, indicating that in a positive learning atmosphere and supportive social environment, social influence can still increase learners' confidence and trust in the platform, thereby increasing their willingness to use the platform. Therefore, educational institutions can indirectly promote platform use by building a supportive social environment to enhance design learners' intrinsic motivation. This respect and support for learners' individual needs can help reduce educational gaps, especially among learners with fewer resources. Virtual platforms can provide a more equal learning experience, thereby improving educational equity.

Secondly, the positive effect of rewards on behavioral intention further supports the research of Bai and Xiao et al. [129,130], indicating that appropriate reward mechanisms can effectively motivate learners to use the platform. Through points, scholarships, or recognition certificates, educational institutions can effectively stimulate students' motivation to participate. Similarly, the positive effect of teacher support on behavioral intention verifies the research conclusions of Fryer and Domen et al. [106,108], highlighting the key role of teachers in technology promotion. Educational institutions can enhance students' willingness to use the platform by training and guiding teachers to ensure that teachers actively promote and demonstrate the use of AI collaboration platforms in the classroom. The positive effect of convenience conditions on behavioral intention supports the findings of Teng et al. [57], indicating that the platform's technology and resource usability are crucial in improving the willingness to use the platform. Therefore, platform designers should continue to optimize the user experience to meet learners' technical needs and enhance their willingness to use it. In particular, for students who lack convenient resources, the convenience of the educational metaverse platform can significantly narrow the gap between them and resource-rich students, thereby providing an effective way to achieve educational equity.

Thirdly, this study confirms the key influence of cognitive factors on behavioral intention, providing practical guidance for designing learners and educational institutions. For example, the significant effect of self-efficacy on behavioral intention indicates that when learners' confidence in their ability to use technology increases, their willingness to use the platform also increases [131]. Educational institutions can help students build confidence by providing technical training and practice opportunities, especially in the early stages of learning. Similarly, the positive effects of outcome expectations and trust further emphasize the key role of platform reliability and data security in improving platform usage intention [132]. Educational institutions and platform developers can enhance students' trust in the platform by demonstrating the effectiveness of the platform and making clear data privacy commitments.

In addition, the finding that external factors indirectly affect behavioral intention through cognitive factors provides strong support for the promotion strategies of the educational metaverse and AI platforms. First, social influence mediates behavioral intention through self-efficacy and trust, which shows that in the educational metaverse environment, although social support and peer influence do not directly affect learners' intention to use a platform, they can indirectly enhance their willingness to use it by improving their self-efficacy and trust. Based on this, when designing promotion strategies, educational institutions should pay more attention to how to establish a supportive learning community and enable learners to feel the indirect influence of social support when using the platform through trust building and self-efficacy improvement. Second, rewards had a significant direct effect on behavioral intention, but the indirect effect through self-efficacy was relatively weak. This suggests that in reward design, educational institutions or platforms should avoid overreliance on indirect channels (such as only improving self-efficacy) and should more directly motivate user participation. For example, through immediate feedback and clear reward mechanisms, learners can directly obtain positive incentives during use, thereby effectively mobilizing their learning enthusiasm. Third, the mediating effect of teacher support on behavioral intention was particularly reflected in the self-efficacy path. This further highlights the important role of teachers in technology promotion. Educational institutions can provide targeted training for teachers so that they can effectively support students in the use of the platform and help students overcome technical barriers in the process of use, thereby significantly improving learners' self-confidence and willingness to continue using it. Finally, the mediating role of trust is significant in multiple paths. This shows that in the process of platform promotion, the establishment of trust is a key factor which cannot be ignored. Educational institutions and platform designers should attach

importance to data privacy and security and build user trust through transparent privacy policies and strict data protection measures.

This study provides important practical value for design learners and educational institutions in terms of technology acceptance and application. As the application of educational metaverse and AI collaborative platforms in future design education becomes increasingly common, the results of this study provide empirical support for how to effectively promote students' willingness to use them. By enhancing learners' self-efficacy and trust, educational institutions can more effectively improve students' acceptance of emerging technologies. At the same time, the findings of this study emphasize the potential of the educational metaverse in reducing educational resource gaps and improving educational equity. Compared with previous studies, this study further reveals how external factors and cognitive factors directly or indirectly affect behavioral intentions through a more comprehensive multi-level analysis and provides specific practical suggestions and strategic support for the promotion of educational metaverse and AI platforms so as to better achieve the goal of educational equity.

7. Implications and Limitations

7.1. Implications

In terms of theoretical contributions, this study offers several key insights. First, this research applied social cognitive theory (SCT) for the first time to analyze design learners' behavioral intentions within the context of educational metaverse and AI collaboration (EMH-AIc) platforms. This approach expands the application of SCT beyond traditional settings, providing a new theoretical framework for examining user behavior in design education and virtual environments. By integrating the three dimensions of the external environment, cognitive factors, and behavioral intention, this study fills a significant theoretical gap regarding design learning in the educational metaverse. Second, the theoretical model developed in the study effectively revealed the mechanisms behind learners' behavioral intentions on these platforms, offering a solid theoretical foundation for future research on sustained user engagement with educational metaverse and AI platforms. Furthermore, this study highlights how external factors (such as rewards and teacher support) and cognitive factors (like self-efficacy and trust) directly and indirectly influence design learners' usage behaviors. This deepens the understanding of how these factors affect decision-making processes. Lastly, the findings provide new empirical support for the broader academic field, particularly regarding the complexities of behavioral intention formation in technology platforms. These contributions help advance the understanding of user behavior in cutting-edge educational technologies.

From a practical standpoint, this study offers valuable insights for educational metaverse platform developers and administrators. First, this research highlights that enhancing the reward system and teacher support significantly boosts user engagement. Thus, developers should focus on designing more incentive mechanisms, such as real-time feedback on learning outcomes or structured reward systems, to motivate learners' sustained usage. Second, self-efficacy and trust have been identified as critical factors driving platform use. Strengthening users' confidence in their ability to use the platform while ensuring security and data privacy will enhance the platform's attractiveness and user retention. Lastly, this study emphasizes the importance of optimizing technical resources and user experience by providing seamless technical support to ensure the platform's long-term development and user retention. These recommendations offer clear directions for improving educational metaverse platforms and contributing to fostering continuous innovation in the educational technology sector.

7.2. Limitations and Future Research Directions

While this study provides theoretical support for understanding the behavioral intentions of design learners to use the educational metaverse and AI collaboration platform, it also has some limitations. First, the sample size of this study was relatively limited, mainly focusing on design learners in a specific geographical area. This limited sample may limit the extrapolation and generalizability of the results and make it difficult to fully represent learners from different cultural backgrounds or geographical areas. This limitation may limit the applicability of the research results globally, especially for regions with other educational systems or design traditions. Second, this study used cross-sectional data analysis, which made it difficult to capture the dynamic changes in learners' behavioral intentions over time and therefore insufficient to fully explain the motivations and behaviors of design learners when using the platform for a long time. In addition, although the PLS-SEM analysis method effectively revealed the relationship between variables, its causal inference ability was weak and failed to deeply explore the causal mechanism between external factors and behavioral intentions. At the same time, this study did not fully evaluate the practical application utility of complex technologies (such as virtual reality and augmented reality) in the educational metaverse, and thus the discussion on the role of these technologies in improving learning experiences and outcomes was relatively limited.

Based on the above limitations, future research should be improved in many aspects. First, the research should expand the sample size and increase geographical and cultural diversity to enhance the general applicability of the research results. Including design learners from different regions and educational backgrounds in the study can better verify the universality of the research conclusions. Secondly, it is recommended to adopt a longitudinal research design to continuously track learners' behaviors and feedback on the educational metaverse platform, as well as deeply explore the temporal changes and driving factors of behavioral intentions. In addition, future research can consider using experimental designs or causal inference methods to more accurately clarify the causal relationship between external factors, cognitive factors, and behavioral intentions. Finally, it is recommended that future research further explore the application value of technologies such as virtual reality and augmented reality in the educational metaverse and evaluate the actual improvement effects of these technologies on learning experience and learning outcomes. These improvements will provide more solid theoretical and empirical support for the sustainable development and innovative application of the educational metaverse platform.

8. Conclusions

This study, through the measurement and validation of a multidimensional model, addressed three core research questions and identified the key factors influencing design learners' use of the educational metaverse for human-AI collaboration (EMH-AIc). Specifically, this research highlights the significance of external factors like rewards and facilitating conditions in driving learners' behavioral intentions, while revealing a limited role for social influence. This finding suggests that educational policy should emphasize the development of platforms which cater to learners' intrinsic motivation and personalized needs rather than relying primarily on peer pressure or social expectations. Teacher support emerged as an essential driver, underscoring the need for educational institutions to equip teachers with tools and training to facilitate effective platform integration. These insights provide not only a theoretical basis for understanding design learners' intentions but also practical guidance for platform design, management, and training frameworks in educational settings.

Moreover, this study confirms the central role of cognitive factors such as self-efficacy, outcome expectations, and trust in shaping behavioral intentions. In particular, self-efficacy and trust acted as critical mediators through which external factors indirectly influenced behavioral intentions. This finding suggests that educational policymakers and platform designers should prioritize initiatives which build learner confidence and establish trust in virtual environments, as these aspects are pivotal to fostering sustained positive engagement. For instance, enhancing data security measures and transparently communicating privacy policies can improve trust, while providing accessible training can boost self-

efficacy. These strategies are instrumental in ensuring that learners' initial interest in EMH-AIc platforms translates into long-term, active use.

In addition to informing educational practices within design education, this study opens avenues for exploring EMH-AIc applications in broader educational contexts. Future research should consider adapting and extending this model to other disciplines, particularly those that involve collaborative or project-based learning, such as engineering, business, and the arts. Longitudinal studies could also shed light on how learners' behavioral intentions and interaction patterns evolve over time, thereby providing insights for iterative improvement of the educational metaverse. Furthermore, experimental designs which assess the causal impact of specific features—such as real-time feedback and personalized AI guidance—on learning outcomes could enhance our understanding of EMH-AIc's efficacy across diverse educational fields. These directions will contribute to building a more comprehensive framework for human-AI collaboration in education, aligning with the evolving needs of modern learners and supporting the development of inclusive, adaptable, and innovative learning ecosystems.

Author Contributions: Conceptualization, R.W. and Y.P.; methodology, R.W. and L.G.; software, Q.H.; validation, R.W., J.L. and Q.H.; formal analysis, Q.H.; investigation, R.W.; writing—original draft preparation, R.W.; writing—review and editing, R.W., J.L. and L.G.; visualization, R.W.; supervision, Y.P. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: All subjects gave their informed consent for inclusion before they participated in this study. Ethics approval is not required for this type of study. This study was conducted while following the local legislation (https://www.law.go.kr/LSW//lsLinkCommonInfo. do?lspttninfSeq=75929&chrClsCd=010202, accessed on 21 August 2024).

Informed Consent Statement: Informed consent was obtained from all subjects involved in this study.

Data Availability Statement: All data generated or analyzed during this study are included in this article. The raw data are available from the corresponding author upon reasonable request.

Acknowledgments: The authors would like to thank all the participants in this study for their time and willingness to share their experiences and feelings.

Conflicts of Interest: The authors declare no conflicts of interest.

Variables	Items and Issues	Factor Loads	References
	SIE1: People who influence me believe I should use the educational metaverse for AI collaboration in learning.	0.812	
Social influence (SIE)	SIE2: The people important to me support my use of the educational metaverse for AI collaboration in learning.	0.809	[00 100 100]
(4 items)	SIE3: I plan to use the educational metaverse for AI collaboration because others are using it.	0.819	[83,102,103]
	SIE4: I want to know if my performance in the educational metaverse will leave a good impression on my family, teachers, or friends.	0.816	
	RW1: Receiving rewards in the educational metaverse makes me feel recognized for my hard work.	0.83	
Rewards (RW)	RW2: I believe my efforts in the educational metaverse will be rewarded.	0.832	
(4 items)	RW3: I might receive extra points, badges, or verbal recognition for using the educational metaverse.	0.837	[80,104,105]
	RW4: I enjoy the reward system in the educational metaverse, and I think it is suitable for the platform.	0.83	

Appendix A. Research Constructs and Factor Loading

Teacher support (TS) (3 items)	TS1: My teacher has provided me with many options on how to complete the assignments.	0.856	[106–109]
	TS2: From the beginning, my teacher has actively sparked our curiosity and interest in using the educational metaverse for AI collaboration.	0.88	
	TS3: My teacher has clearly explained the importance of using the educational metaverse for AI collaboration in learning.	0.838	
Facilitating conditions (FC) (4 items)	FC1: I have the resources necessary to use the educational metaverse.	0.845	[83,97,110]
	FC2: I know how to use the educational metaverse for AI collaboration in learning.	0.788	
	FC3: When I face difficulties in the educational metaverse, designated individuals or groups are available to help.	0.826	
	FC4: I have easy access to the materials I need to develop educational activities through mobile devices.	0.832	
Self-efficacy (SE) (5 items)	SE1: If I want to, I can easily learn through AI collaboration in the educational metaverse.	0.838	[63,75,86,103,111]
	SE2: I am confident in my understanding of the functions and content of the educational metaverse for AI	0.795	
	SE3: Even if no one shows me how to use the educational metaverse for AI collaboration, I am confident I can use it.	0.802	
	SE4: If I want to study through the educational metaverse, it is definitely feasible for me.	0.814	
	SE5: Whether or not I use the educational metaverse for learning depends mostly on myself.	0.823	
Outcome expectation (OE) (4 items)	OE1: If I use the educational metaverse for AI collaboration, my learning efficiency will improve.	0.856	[86,100,103,111]
	OE2: Using the educational metaverse for AI collaboration will increase the quality of my output.	0.834	
	OE3: Important people in my life (such as family, teachers, or friends) would support me in using the educational metaverse for AI collaboration to improve my learning outcomes.	0.847	
	OE4: Using the educational metaverse for AI collaboration is useful for my learning.	0.825	
Trust (TRU) (4 items)	TRU1: The educational metaverse provides reliable resources for design education.	0.798	
	TRU2: The educational information obtained through AI collaboration in the educational metaverse is safe and	0.844	[63,112]
	TRU3: The people I interact with on the educational metaverse platform are trustworthy.	0.84	
	TRU4: Learning on this platform is safe and trustworthy.	0.843	
Behavioral intention (BI) (4 items)	BI1: I intend to use the educational metaverse for AI collaboration in my studies.	0.808	[63,83,102,103,110]
	BI2: I expect to continue using the educational metaverse	0.847	
	I control for Al collaboration in learning. BI3: I plan to regularly use the educational metaverse for Al collaboration in both learning and work in the future	0.833	
	BI4: I think using the educational metaverse for AI collaboration is necessary to meet my learning needs.	0.829	

References

- 1. United Nations. Recognizing and Overcoming Inequity in Education. Available online: https://www.un.org/en/un-chronicle/ recognizing-and-overcoming-inequity-education (accessed on 22 May 2024).
- Martin United Nations Sustainable Development Goals—Time for Global Action for People and Planet. United Nations Sustainable Development. Available online: https://www.un.org/sustainabledevelopment/education/ (accessed on 16 September 2024).
- 3. Erlangga, D.T. Student problems in online learning: Solutions to keep education going on. *J. Engl. Lang. Teach. Learn.* **2022**, *3*, 21–26. [CrossRef]
- 4. Wang, S. Smart Education—The necessity and prospect of big data mining and artificial intelligence technology in art education. *J. Phys. Conf. Ser.* **2020**, *1648*, 042060. [CrossRef]
- Chen, X.; Zhong, Z.; Wu, D. Metaverse for Education: Technical Framework and Design Criteria. *IEEE Trans. Learn. Technol.* 2023, 16, 1034–1044. [CrossRef]
- Kye, B.; Han, N.; Kim, E.; Park, Y.; Jo, S. Educational Applications of Metaverse: Possibilities and Limitations. J. Educ. Eval. Health Prof. 2021, 18, 2–8. [CrossRef] [PubMed]
- 7. Mystakidis, S. Metaverse. Encyclopedia 2022, 2, 486–497. [CrossRef]
- VirtuARealities—Google Expeditions. Available online: https://sites.google.com/view/virtuarealities/vr-resources/googleexpeditions (accessed on 19 September 2024).
- 9. Virbela: A Virtual World for Work, Education & Events. Available online: https://www.virbela.com/ (accessed on 19 September 2024).
- 10. Labster | Virtual Labs for Universities and High Schools. Available online: https://www.labster.com/ (accessed on 19 September 2024).
- 11. Class VR Use Cases. Available online: https://www.classvr.com/?ppc_keyword=&gad_source=1&gclid=Cj0KCQiArby5 BhCDARIsAIJvjIRRmzZ50V-XvGY0e-n_haaoBygvO76_HxFImOBDj--VV6Y-Ki6ZnCwaAuF8EALw_wcB (accessed on 19 September 2024).
- 12. Spatial Computing Artificial Intelligence. Available online: https://engagevr.io/ (accessed on 19 September 2024).
- 13. How the Metaverse Can Transform Education. Meta. 2023. Available online: https://about.fb.com/news/2023/04/how-the-metaverse-can-transform-education/ (accessed on 12 April 2024).
- 14. Metaverse Market Share & Growth | Forecast Analysis [2027]. Available online: https://www.fortunebusinessinsights.com/ metaverse-market-106574 (accessed on 16 May 2024).
- 15. World Leader in AI Computing. Available online: https://www.nvidia.com/en-us/ (accessed on 16 September 2024).
- 16. Metaverse Education—Worldwide | Statista Market Forecast. Available online: https://www.statista.com/outlook/amo/ metaverse/metaverse-education/worldwide (accessed on 16 September 2024).
- 17. The Business Research Company. Metaverse in Education Global Market Report 2024. Available online: https://www.researchandmarkets.com/reports/5931030/metaverse-in-education-global-market-report (accessed on 15 September 2024).
- 18. Olatunde-Aiyedun, T.G. Artificial Intelligence (AI) in Education: Integration of AI Into Science Education Curriculum in Nigerian Universities; University of Abuja: Abuja, Nigeria, 2024.
- Abulibdeh, A.; Zaidan, E.; Abulibdeh, R. Navigating the Confluence of Artificial Intelligence and Education for Sustainable Development in the Era of Industry 4.0: Challenges, Opportunities, and Ethical Dimensions. J. Clean. Prod. 2024, 437, 140527. [CrossRef]
- 20. Elbanna, S.; Armstrong, L. Exploring the Integration of ChatGPT in Education: Adapting for the Future. *Manag. Sustain. Arab Rev.* **2024**, *3*, 16–29. [CrossRef]
- 21. Exploring the Future of Graphic Design in the Metaverse. Available online: https://www.berlinsbi.com/blog/exploring-the-future-of-graphic-design-in-the-metaverse (accessed on 16 October 2024).
- 22. How the Metaverse Is Having an Impact on the Fashion Industry? Available online: https://power.xyz/gaming/story/impactmetaverse-fashion-industry/ (accessed on 16 October 2024).
- 23. Wu, J.; Gao, G. Edu-Metaverse: Internet Education Form with Fusion of Virtual and Reality; Atlantis Press: Amsterdam, The Netherlands, 2022; pp. 1082–1085.
- 24. Wang, M.; Yu, H.; Bell, Z.; Chu, X. Constructing an Edu-Metaverse Ecosystem: A New and Innovative Framework. *IEEE Trans. Learn. Technol.* **2022**, *15*, 685–696. [CrossRef]
- 25. Chen, Z. Exploring the Application Scenarios and Issues Facing Metaverse Technology in Education. *Interact. Learn. Environ.* **2024**, *32*, 1975–1987. [CrossRef]
- 26. De Felice, F.; Petrillo, A.; Iovine, G.; Salzano, C.; Baffo, I. How Does the Metaverse Shape Education? A Systematic Literature Review. *Appl. Sci.* 2023, *13*, 5682. [CrossRef]
- 27. Education 2030: Incheon Declaration and Framework for Action Towards Inclusive and Equitable Quality Education and Lifelong Learning for All—UNESCO IITE. Available online: https://iite.unesco.org/publications/education-2030-incheon-declaration-framework-action-towards-inclusive-equitable-quality-education-lifelong-learning/ (accessed on 16 October 2024).
- Ktoridou, D.; Epaminonda, E.; Efthymiou, L.; Michailidis, M. Exploring Metaverse Learning Knowledge and Acceptance: A Study of Students' Opinions in Higher Education. In Proceedings of the 2024 IEEE Global Engineering Education Conference (EDUCON), Kos, Greece, 8–11 May 2024; pp. 1–6.

- 29. Bandura, A. Social Cognitive Theory: An Agentic Perspective. Annu. Rev. Psychol. 2001, 52, 1–26. [CrossRef] [PubMed]
- Bandura, A. Social cognitive theory of mass communication. In *Media Effects*; Routledge: Oxfordshire, UK, 2008; ISBN 978-0-203-87711-1.
- Dionisio, J.D.N.; Burns, W.G., III; Gilbert, R. 3D Virtual Worlds and the Metaverse: Current Status and Future Possibilities. ACM Comput. Surv. 2013, 45, 31–34. [CrossRef]
- Ludlow, P.; Wallace, M. The Second Life Herald: The Virtual Tabloid That Witnessed the Dawn of the Metaverse; MIT Press: Cambridge, MA, USA, 2007; ISBN 978-0-262-12294-8.
- 33. Mistretta, S. The Metaverse—An Alternative Education Space. AI Comput. Sci. Robot. Technol. 2022, 1–23. [CrossRef]
- Shoaib, M.; Sayed, N.; Singh, J.; Shafi, J.; Khan, S.; Ali, F. AI Student Success Predictor: Enhancing Personalized Learning in Campus Management Systems. *Comput. Human. Behav.* 2024, 158, 108301. [CrossRef]
- 35. Alqahtani, T.; Badreldin, H.A.; Alrashed, M.; Alshaya, A.I.; Alghamdi, S.S.; bin Saleh, K.; Alowais, S.A.; Alshaya, O.A.; Rahman, I.; Al Yami, M.S.; et al. The Emergent Role of Artificial Intelligence, Natural Learning Processing, and Large Language Models in Higher Education and Research. *Res. Soc. Adm. Pharm.* 2023, 19, 1236–1242. [CrossRef]
- 36. Shaik, T.; Tao, X.; Li, Y.; Dann, C.; McDonald, J.; Redmond, P.; Galligan, L. A Review of the Trends and Challenges in Adopting Natural Language Processing Methods for Education Feedback Analysis. *IEEE Access* **2022**, *10*, 56720–56739. [CrossRef]
- 37. Chen, L.; Chen, P.; Lin, Z. Artificial Intelligence in Education: A Review. IEEE Access 2020, 8, 75264–75278. [CrossRef]
- Alhothali, A.; Albsisi, M.; Assalahi, H.; Aldosemani, T. Predicting Student Outcomes in Online Courses Using Machine Learning Techniques: A Review. Sustainability 2022, 14, 6199. [CrossRef]
- 39. Rospigliosi, P.A. Adopting the Metaverse for Learning Environments Means More Use of Deep Learning Artificial Intelligence: This Presents Challenges and Problems. *Interact. Learn. Environ.* **2022**, *30*, 1573–1576. [CrossRef]
- 40. Saqr, R.R.; Al-Somali, S.A.; Sarhan, M.Y. Exploring the Acceptance and User Satisfaction of AI-Driven e-Learning Platforms (Blackboard, Moodle, Edmodo, Coursera and edX): An Integrated Technology Model. *Sustainability* **2024**, *16*, 204. [CrossRef]
- 41. Bhutoria, A. Personalized Education and Artificial Intelligence in the United States, China, and India: A systematic review using a human-in-the-loop model. *Comput. Educ. Artif. Intell.* **2022**, *3*, 100068. [CrossRef]
- 42. Fan, X.; Zhong, X. Artificial Intelligence-Based Creative Thinking Skill Analysis Model Using Human–Computer Interaction in Art Design Teaching. *Comput. Electr. Eng.* 2022, 100, 107957. [CrossRef]
- 43. Kim, J.; Lee, H.; Cho, Y.H. Learning Design to Support Student-AI Collaboration: Perspectives of Leading Teachers for AI in Education. *Educ. Inf. Technol.* 2022, 27, 6069–6104. [CrossRef]
- Walmsley, K. Cutting-Edge Practical Research on Generative Design, IoT and Digital Twins. In *Innovation in Construction: A Practical Guide to Transforming the Construction Industry*; Ghaffar, S.H., Mullett, P., Pei, E., Roberts, J., Eds.; Springer: Cham, Switzerland, 2022; pp. 117–145. ISBN 978-3-030-95798-8.
- 45. Thuy, V.T.T. AI technology and the potential applications in architectural design and applied fine arts education in the Vietnam context. In Proceedings of the 7th International Conference on Geotechnics, Civil Engineering and Structures (CIGOS), Ho Chi Minh City, Vietnam, 4–5 April 2024; Springer Nature: Singapore, 2024; pp. 975–987.
- 46. Almulla, M.A. Investigating influencing factors of learning satisfaction in AI ChatGPT for research: University Students Perspective. *Heliyon* **2024**, *10*, e32220. [CrossRef]
- He, C.; Sun, B. Application of Artificial Intelligence Technology in Computer Aided Art Teaching. *Comput.-Aided Design Appl.* 2021, 18, 118–129. [CrossRef]
- Phakamach, P.; Senarith, P.; Wachirawongpaisarn, S. The metaverse in education: The future of immersive teaching & learning. RICE J. Creat. Entrep. Manag. 2022, 3, 75–88. [CrossRef]
- Pan, D. Research on the Construction and Reform of Art Design Education from the Perspective of the Metaverse. *Trans. Comp. Educ.* 2023, 5, 22–27. [CrossRef]
- 50. Jin, Y.; Tiejun, Z. The Application of Metaverse XiRang Game in the Mixed Teaching of Art and Design in Colleges and Universities. *Educ. Inf. Technol.* **2023**, *28*, 15625–15655. [CrossRef]
- 51. Chua, H.W.; Yu, Z. A Systematic Literature Review of the Acceptability of the Use of Metaverse in Education over 16 Years. *J. Comput. Educ.* 2024, *11*, 615–665. [CrossRef]
- 52. Yang, F.; Ren, L.; Gu, C. A Study of College Students' Intention to Use Metaverse Technology for Basketball Learning Based on UTAUT2. *Heliyon* 2022, *8*, e10562. [CrossRef]
- 53. Chen, X.; Zou, D.; Xie, H.; Wang, F.L. Metaverse in Education: Contributors, Cooperations, and Research Themes. *IEEE Trans. Learn. Technol.* 2023, *16*, 1111–1129. [CrossRef]
- 54. Wang, G.; Shin, C. Influencing Factors of Usage Intention of Metaverse Education Application Platform: Empirical Evidence Based on PPM and TAM Models. *Sustainability* **2022**, *14*, 17037. [CrossRef]
- Akour, I.A.; Al-Maroof, R.S.; Alfaisal, R.; Salloum, S.A. A Conceptual Framework for Determining Metaverse Adoption in Higher Institutions of Gulf Area: An Empirical Study Using Hybrid SEM-ANN Approach. *Comput. Educ. Artif. Intell.* 2022, 3, 100052. [CrossRef]
- AlHamad, A.Q.; Alomari, K.M.; Alshurideh, M.; Al Kurdi, B.; Salloum, S.; Al-Hamad, A.Q. The Adoption of Metaverse Systems: A Hybrid SEM—ML Method. In Proceedings of the 2022 International Conference on Electrical, Computer, Communications and Mechatronics Engineering (ICECCME), Malé, Maldives, 16–18 November 2022; pp. 1–5.

- 57. Teng, Z.; Cai, Y.; Gao, Y.; Zhang, X.; Li, X. Factors Affecting Learners' Adoption of an Educational Metaverse Platform: An Empirical Study Based on an Extended UTAUT Model. *Mob. Inf. Syst.* **2022**, 2022, 5479215. [CrossRef]
- 58. Al-Adwan, A.S.; Al-Debei, M.M. The Determinants of Gen Z's Metaverse Adoption Decisions in Higher Education: Integrating UTAUT2 with Personal Innovativeness in IT. *Educ. Inf. Technol.* **2024**, *29*, 7413–7445. [CrossRef]
- Lo Schiavo, M.; Prinari, B.; Saito, I.; Shoji, K.; Benight, C.C. A Dynamical Systems Approach to Triadic Reciprocal Determinism of Social Cognitive Theory. *Math. Comput. Simul.* 2019, 159, 18–38. [CrossRef]
- 60. Bandura, A. Human Agency in Social Cognitive Theory. Am. Psychol. 1989, 44, 1175–1184. [CrossRef]
- 61. Ryan, R. The Oxford Handbook of Human Motivation; Oxford University Press: Oxford, UK, 2019; ISBN 978-0-19-066646-0.
- 62. Compeau, D.R.; Higgins, C.A. Computer Self-Efficacy: Development of a Measure and Initial Test. *MIS Q.* **1995**, *19*, 189–211. [CrossRef]
- 63. Wu, D.; Gu, H.; Gu, S.; You, H. Individual Motivation and Social Influence: A Study of Telemedicine Adoption in China Based on Social Cognitive Theory. *Health Policy Technol.* **2021**, *10*, 100525. [CrossRef]
- 64. Lin, C.-P. Learning Virtual Community Loyalty Behavior from a Perspective of Social Cognitive Theory. *Int. J. Hum. Comput. Interact.* **2010**, *26*, 345–360. [CrossRef]
- Lim, J.S.; Choe, M.-J.; Zhang, J.; Noh, G.-Y. The Role of Wishful Identification, Emotional Engagement, and Parasocial Relationships in Repeated Viewing of Live-Streaming Games: A Social Cognitive Theory Perspective. *Comput. Hum. Behav.* 2020, 108, 106327. [CrossRef]
- Woodcock, S.; Tournaki, N. Bandura's Triadic Reciprocal Determinism Model and Teacher Self-Efficacy Scales: A Revisit. *Teach. Dev.* 2023, 27, 75–91. [CrossRef]
- 67. Li, K.; Wijaya, T.T.; Chen, X.; Harahap, M.S. Exploring the Factors Affecting Elementary Mathematics Teachers' Innovative Behavior: An Integration of Social Cognitive Theory. *Sci. Rep.* **2024**, *14*, 2108. [CrossRef] [PubMed]
- Almulla, M.A.; Al-Rahmi, W.M. Integrated Social Cognitive Theory with Learning Input Factors: The Effects of Problem-Solving Skills and Critical Thinking Skills on Learning Performance Sustainability. *Sustainability* 2023, 15, 3978. [CrossRef]
- 69. Devi, B.; Khandelwal, D.B.; Das, M. Application of Bandura's Social Cognitive Theory in the Technology Enhanced, Blended Learning Environment. *Int. J. Appl. Res.* 2017, *3*, 721–724.
- Al-Dokhny, A.; Drwish, A.; Alyoussef, I.; Al-Abdullatif, A. Students' Intentions to Use Distance Education Platforms: An Investigation into Expanding the Technology Acceptance Model through Social Cognitive Theory. *Electronics* 2021, 10, 2992. [CrossRef]
- 71. Almogren, A.S.; Aljammaz, N.A. The Integrated Social Cognitive Theory with the TAM Model: The Impact of M-Learning in King Saud University Art Education. *Front. Psychol.* **2022**, *13*, 1050532. [CrossRef]
- Kan, M.P.H.; Fabrigar, L.R. Theory of Planned Behavior. In *Encyclopedia of Personality and Individual Differences*; Zeigler-Hill, V., Shackelford, T.K., Eds.; Springer: Cham, Switzerland, 2020; pp. 5476–5483. ISBN 978-3-319-24612-3.
- 73. Rasoolimanesh, S.M.; Jaafar, M.; Kock, N.; Ramayah, T. A Revised Framework of Social Exchange Theory to Investigate the Factors Influencing Residents' Perceptions. *Tour. Manag. Perspect.* **2015**, *16*, 335–345. [CrossRef]
- 74. Yamak, S.; Nielsen, S.; Escribá-Esteve, A. The Role of External Environment in Upper Echelons Theory: A Review of Existing Literature and Future Research Directions. *Group Organ. Manag.* **2014**, *39*, 69–109. [CrossRef]
- 75. Chang, C.-T.; Hajiyev, J.; Su, C.-R. Examining the Students' Behavioral Intention to Use e-Learning in Azerbaijan? The General Extended Technology Acceptance Model for E-Learning Approach. *Comput. Educ.* **2017**, *111*, 128–143. [CrossRef]
- Rubenstein, L.D.; Ridgley, L.M.; Callan, G.L.; Karami, S.; Ehlinger, J. How Teachers Perceive Factors That Influence Creativity Development: Applying a Social Cognitive Theory Perspective. *Teach. Teach. Educ.* 2018, 70, 100–110. [CrossRef]
- Lakhal, S.; Khechine, H. Student Intention to Use Desktop Web-Conferencing According to Course Delivery Modes in Higher Education. Int. J. Manag. Educ. 2016, 14, 146–160. [CrossRef]
- Salloum, S.A.; Shaalan, K. Factors Affecting Students' Acceptance of E-Learning System in Higher Education Using UTAUT and Structural Equation Modeling Approaches. In Proceedings of the International Conference on Advanced Intelligent Systems and Informatics, Cairo, Egypt, 1–3 September 2018; Hassanien, A.E., Tolba, M.F., Shaalan, K., Azar, A.T., Eds.; Springer: Cham, Switzerland, 2019; pp. 469–480.
- 79. Lawler, E.J. An Affect Theory of Social Exchange. Am. J. Sociol. 2001, 107, 321–352. [CrossRef]
- 80. Labsomboonsiri, S.; Mathews, S.; Luck, E.; Mulcahy, R. Recognise Me or Pay Me? How Rewards for Online Restaurant Reviews Impact Continuity: A Cross-Country Investigation. *Int. J. Hosp. Manag.* **2022**, *106*, 103289. [CrossRef]
- Granziera, H.; Liem, G.A.D.; Chong, W.H.; Martin, A.J.; Collie, R.J.; Bishop, M.; Tynan, L. The Role of Teachers' Instrumental and Emotional Support in Students' Academic Buoyancy, Engagement, and Academic Skills: A Study of High School and Elementary School Students in Different National Contexts. *Learn. Instr.* 2022, 80, 101619. [CrossRef]
- 82. Sakiz, G.; Pape, S.J.; Hoy, A.W. Does Perceived Teacher Affective Support Matter for Middle School Students in Mathematics Classrooms? *J. Sch. Psychol.* 2012, *50*, 235–255. [CrossRef]
- Nikou, S.A.; Economides, A.A. Mobile-Based Assessment: Investigating the Factors That Influence Behavioral Intention to Use. Comput. Educ. 2017, 109, 56–73. [CrossRef]
- 84. Bandura, A. Social Cognitive Theory of Self-Regulation. Organ. Behav. Human. Decis. Process. 1991, 50, 248–287. [CrossRef]
- 85. Schunk, D.H.; DiBenedetto, M.K. Motivation and Social Cognitive Theory. Contemp. Educ. Psychol. 2020, 60, 101832. [CrossRef]

- Cai, Y.; Shi, W. The Influence of the Community Climate on Users' Knowledge-Sharing Intention: The Social Cognitive Theory Perspective. *Behav. Inf. Technol.* 2022, 41, 307–323. [CrossRef]
- 87. Bandura, A.; Adams, N.E. Analysis of Self-Efficacy Theory of Behavioral Change. Cogn. Ther. Res. 1977, 1, 287–310. [CrossRef]
- 88. Zhang, T.; Tao, D.; Qu, X.; Zhang, X.; Zeng, J.; Zhu, H.; Zhu, H. Automated Vehicle Acceptance in China: Social Influence and Initial Trust Are Key Determinants. *Transp. Res. Part. C Emerg. Technol.* **2020**, *112*, 220–233. [CrossRef]
- Howard, J.L.; Bureau, J.S.; Guay, F.; Jane, X.Y.C.; Ryan, R.M. Student Motivation and Associated Outcomes: A Meta-Analysis from Self-Determination Theory. 2021. Available online: https://journals.sagepub.com/doi/abs/10.1177/1745691620966789 (accessed on 29 October 2024).
- 90. Malek, S.L.; Sarin, S.; Haon, C. Extrinsic Rewards, Intrinsic Motivation, and New Product Development Performance. J. Prod. Innov. Manag. 2020, 37, 528–551. [CrossRef]
- Shneor, R.; Mrzygłód, U.; Adamska-Mieruszewska, J.; Fornalska-Skurczyńska, A. The Role of Social Trust in Reward Crowdfunding Campaigns' Design and Success. *Electron. Mark.* 2022, 32, 1103–1118. [CrossRef]
- 92. Mou, Y.; Ma, Y.; Guo, D.; Ding, Z. Effect of Platform Gamification Rewards on User Stickiness. In *Management Decision*; Emerald Publishing Limited: Leeds, UK, 2024; *ahead-of-print*. [CrossRef]
- Öqvist, A.; Malmström, M. What Motivates Students? A Study on the Effects of Teacher Leadership and Students' Self-Efficacy. Int. J. Leadersh. Educ. 2018, 21, 155–175. [CrossRef]
- 94. Maclellan, E. How Might Teachers Enable Learner Self-Confidence? A Review Study. Educ. Rev. 2014, 66, 59–74. [CrossRef]
- 95. Platz, M. Trust Between Teacher and Student in Academic Education at School. J. Philos. Educ. 2021, 55, 688–697. [CrossRef]
- 96. Van Maele, D.; Van Houtte, M.; Forsyth, P.B. Introduction: Trust as a Matter of Equity and Excellence in Education. In *Trust and School Life: The Role of Trust for Learning, Teaching, Leading, and Bridging*; Van Maele, D., Forsyth, P.B., Van Houtte, M., Eds.; Springer: Dordrecht, The Netherlands, 2014; pp. 1–33. ISBN 978-94-017-8014-8.
- 97. Sánchez-Prieto, J.C.; Olmos-Migueláñez, S.; García-Peñalvo, F.J. Informal Tools in Formal Contexts: Development of a Model to Assess the Acceptance of Mobile Technologies among Teachers. *Comput. Human. Behav.* **2016**, *55*, 519–528. [CrossRef]
- Ozturk, A.B.; Bilgihan, A.; Nusair, K.; Okumus, F. What Keeps the Mobile Hotel Booking Users Loyal? Investigating the Roles of Self-Efficacy, Compatibility, Perceived Ease of Use, and Perceived Convenience. Int. J. Inf. Manag. 2016, 36, 1350–1359. [CrossRef]
- 99. Bang, H.-K.; Ellinger, A.E.; Hadjimarcou, J.; Traichal, P.A. Consumer Concern, Knowledge, Belief, and Attitude toward Renewable Energy: An Application of the Reasoned Action Theory. *Psychol. Mark.* 2000, *17*, 449–468. [CrossRef]
- Bao, Z.; Han, Z. What Drives Users' Participation in Online Social Q&A Communities? An Empirical Study Based on Social Cognitive Theory. *Aslib J. Inf. Manag.* 2019, 71, 637–656. [CrossRef]
- 101. Chiu, C.-M.; Hsu, M.-H.; Wang, E.T.G. Understanding Knowledge Sharing in Virtual Communities: An Integration of Social Capital and Social Cognitive Theories. *Decis. Support Syst.* **2006**, *42*, 1872–1888. [CrossRef]
- 102. Li, C.-Y. Persuasive Messages on Information System Acceptance: A Theoretical Extension of Elaboration Likelihood Model and Social Influence Theory. *Comput. Human. Behav.* 2013, 29, 264–275. [CrossRef]
- Citizen's Adoption of an e-Government System: Validating Extended Social Cognitive Theory (SCT). Gov. Inf. Q. 2015, 32, 172–181.
 [CrossRef]
- 104. Peng, X.; Xing, Y.; Tian, Y.; Fei, M.; Wang, Q. Nonmonetary Rewards of Referral Reward Programs and Recommendation Intention: The Role of Reward–Product Congruity. *Decis. Support Syst.* **2023**, *173*, 113999. [CrossRef]
- 105. Cui, X.; Yang, X.; Ren, J.; Lowry, P.B.; Du, T.C. Enhancing Team Creativity among Information Technology Professionals through Knowledge Sharing and Motivational Rewards: A Self-Determination Perspective. In *Data and Information Management*; Elsevier: Amsterdam, The Netherlands, 2024; p. 100075. [CrossRef]
- Domen, J.; Hornstra, L.; Weijers, D.; van der Veen, I.; Peetsma, T. Differentiated Need Support by Teachers: Student-Specific Provision of Autonomy and Structure and Relations with Student Motivation. *Br. J. Educ. Psychol.* 2020, 90, 403–423. [CrossRef]
- 107. Descals-Tomás, A.; Rocabert-Beut, E.; Abellán-Roselló, L.; Gómez-Artiga, A.; Doménech-Betoret, F. Influence of Teacher and Family Support on University Student Motivation and Engagement. *Int. J. Environ. Res. Public. Health* **2021**, *18*, 2606. [CrossRef]
- Fryer, L.K.; Bovee, H.N. Supporting Students' Motivation for e-Learning: Teachers Matter on and offLine. *Internet High. Educ.* 2016, 30, 21–29. [CrossRef]
- Doménech-Betoret, F.; Gómez-Artiga, A.; Abellán-Roselló, L.; Rocabert-Beút, E. MOCSE Centered on Students: Validation of Learning Demands and Teacher Support Scales. Front. Psychol. 2020, 11, 582926. [CrossRef]
- Ain, N.; Kaur, K.; Waheed, M. The Influence of Learning Value on Learning Management System Use: An Extension of UTAUT2. *Inf. Dev.* 2016, 32, 1306–1321. [CrossRef]
- 111. Shahangian, S.A.; Tabesh, M.; Yazdanpanah, M. Psychosocial Determinants of Household Adoption of Water-Efficiency Behaviors in Tehran Capital, Iran: Application of the Social Cognitive Theory. *Urban Clim.* **2021**, *39*, 100935. [CrossRef]
- 112. Barreda, A.A.; Bilgihan, A.; Kageyama, Y. The Role of Trust in Creating Positive Word of Mouth and Behavioral Intentions: The Case of Online Social Networks. *J. Relatsh. Mark.* **2015**, *14*, 16–36. [CrossRef]
- 113. Chanda, R.S.; Pabalkar, V.; Sharma, S. Attitude and Behavioral Intention for Using Metaverse in Education: Learner's Perspective. J. Appl. Res. High. Educ. 2024. ahead-of-print. [CrossRef]
- 114. Parker, C.; Scott, S.; Geddes, A. Snowball Sampling; SAGE Research Methods Foundations: London, UK, 2019.
- 115. Valerio, M.A.; Rodriguez, N.; Winkler, P.; Lopez, J.; Dennison, M.; Liang, Y.; Turner, B.J. Comparing Two Sampling Methods to Engage Hard-to-Reach Communities in Research Priority Setting. *BMC Med. Res. Methodol.* **2016**, *16*, 146. [CrossRef]

- 116. İbili, E.; Ölmez, M.; Cihan, A.; Bilal, F.; İbili, A.B.; Okumus, N.; Billinghurst, M. Investigation of Learners' Behavioral Intentions to Use Metaverse Learning Environment in Higher Education: A Virtual Computer Laboratory. In *Interactive Learning Environments*; Taylor & Francis: Abingdon, UK, 2023.
- 117. Wiangkham, A.; Vongvit, R. Exploring the Drivers for the Adoption of Metaverse Technology in Engineering Education Using PLS-SEM and ANFIS. *Educ. Inf. Technol.* 2024, 29, 7385–7412. [CrossRef]
- 118. Dang, T.-Q.; Tran, P.-T.; Nguyen, L.-T. Are You Ready for Tapping into the Metaverse in Higher Education? Integrated by Dual PLS-SEM and ANN Approach. In *Current and Future Trends on Intelligent Technology Adoption*; Al-Sharafi, M.A., Al-Emran, M., Tan, G.W.-H., Ooi, K.-B., Eds.; Studies in Computational Intelligence; Springer: Cham, Switzerland, 2023; Volume 1128, pp. 63–84. ISBN 978-3-031-48396-7.
- Henseler, J.; Ringle, C.M.; Sarstedt, M. A New Criterion for Assessing Discriminant Validity in Variance-Based Structural Equation Modeling. J. Acad. Mark. Sci. 2015, 43, 115–135. [CrossRef]
- Podsakoff, P.M.; MacKenzie, S.B.; Lee, J.-Y.; Podsakoff, N.P. Common Method Biases in Behavioral Research: A Critical Review of the Literature and Recommended Remedies. J. Appl. Psychol. 2003, 88, 879–903. [CrossRef]
- 121. Akram, K.; Saeed, A.; Bresciani, S.; Rehman, S.U.; Ferraris, A. Factors Affecting Environmental Performance during the Covid-19 Period in the Leather Industry: A Moderated-Mediation Approach. J. Compet. 2022, 14, 5–22. [CrossRef]
- 122. Hair, J.F.; Hult, G.T.M.; Ringle, C.M.; Sarstedt, M. Aprimer on Partial Least Squares Structural Equation Modeling (PLS-SEM). J. Tour. Res. 2014, 6, 211–213.
- 123. Voorhees, C.M.; Brady, M.K.; Calantone, R.; Ramirez, E. Discriminant Validity Testing in Marketing: An Analysis, Causes for Concern, and Proposed Remedies. J. Acad. Mark. Sci. 2016, 44, 119–134. [CrossRef]
- 124. Hamid, M.R.A.; Sami, W.; Sidek, M.H.M. Discriminant Validity Assessment: Use of Fornell & Larcker Criterion versus HTMT Criterion. J. Phys. Conf. Ser. 2017, 890, 012163. [CrossRef]
- 125. Kim, J.H. Multicollinearity and Misleading Statistical Results. Korean J. Anesthesiol. 2019, 72, 558–569. [CrossRef] [PubMed]
- 126. Arain, A.A.; Hussain, Z.; Rizvi, W.H.; Vighio, M.S. Extending UTAUT2 toward Acceptance of Mobile Learning in the Context of Higher Education. *Univ. Access Inf. Soc.* **2019**, *18*, 659–673. [CrossRef]
- 127. Dahri, N.A.; Al-Rahmi, W.M.; Almogren, A.S.; Yahaya, N.; Vighio, M.S.; Al-maatuok, Q.; Al-Rahmi, A.M.; Al-Adwan, A.S. Acceptance of Mobile Learning Technology by Teachers: Influencing Mobile Self-Efficacy and 21st-Century Skills-Based Training. *Sustainability* 2023, 15, 8514. [CrossRef]
- 128. Fouad, A.; Abdel, Z.; Al-Slehat, Z. Exploring the Students' Behavior Intentions to Adopt E-Learning Technology: A Survey Study Based on COVID-19 Crisis. *Int. J. Bus. Manag.* 2021, *16*, 31–41. [CrossRef]
- Xiao, Y.; Hew, K.F.T. Intangible Rewards versus Tangible Rewards in Gamified Online Learning: Which Promotes Student Intrinsic Motivation, Behavioural Engagement, Cognitive Engagement and Learning Performance? Br. J. Educ. Technol. 2024, 55, 297–317. [CrossRef]
- Bai, S.; Gonda, D.E.; Hew, K.F. Effects of Tangible Rewards on Student Learning Performance, Knowledge Construction, and Perception in Fully Online Gamified Learning. In Proceedings of the 2021 IEEE International Conference on Engineering, Technology & Education (TALE), Wuhan, China, 5–8 December 2021; pp. 899–904.
- 131. Maddux, J.E. Self-Efficacy Theory. In Self-Efficacy, Adaptation, and Adjustment: Theory, Research, and Application; Maddux, J.E., Ed.; Springer: Boston, MA, USA, 1995; pp. 3–33. ISBN 978-1-4419-6868-5.
- 132. Li, Y.-J.; Cheung, C.M.K.; Shen, X.-L.; Lee, M.K.O. Promoting Collaborative Learning in Virtual Worlds: The Power of "We". *Inf. Technol. People* **2023**, *36*, 2563–2586. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.